Accelerator Status Report

Run Schedule: March 11, 2010 to August 13, 2010
July 14, 2010
Injector    Energy 61.88 MeV
· QE measurement tool is working now.
· (7/1) The new correctors at aperture A1 were commissioned. Those correctors decouple the interception and steering at A1.
· (7/1) The Wien angle was changed to -73.8 degrees for Hall B.

· QE measurement tool does not work.
Accelerator   Linac Energy 550.00 MeV
· Since Monday day shift we have been performing beam studies because the end stations are not ready for beam. There are about seven shifts available for beam studies and RF recovery. The plan is that Hall B will be ready for beam Thursday (7/15) swing shift.
· The optics has been stable.

· (7/12) False fire alarm. 
· (7/11) CHL tripped (5 hrs.)

· (7/7) The accelerator was turned off from 1315 to 1700 to save power at request of Dominion Power, and beam was restored around midnight.
· (7/6) Scheduled power outage in the East Arc. It went well.
· (7/5) 2L21 liquid level was not stable. It was stabilized by reducing the gradients by 10%.

· (7/4) CHL tripped (5.6 h)

· (7/2) CHL tripped twice (9.1 h)

Hall A---Pass 2, APEX
· The APEX Test runs were done successfully. The hall has been down since Monday (7/12) for installation of DVCS. 

· The APEX Test has been extended for a few.

· The setup was done for APEX Test. MBM1H05H and the Septum magnet is FSD interlocked. The temperatures of the permanent are monitored in the Alarm Handler and CHIME.
· (7/3) Hall A Moller quads went to zero set point at 1521. An Ops-PR was made.

Hall B ---Pass 5, g9-Frost from March 11 to August 13, 2010
· Hall B has been down since July 8 due to issues with its target. 

            The hall is planning to take tomorrow swing shift (7/15) with new energy (pass needs to 
            be changed from 5 to 2). 

· Hall B had problems with its Torus magnet for three days. Now they are having problems

      with its target and will be down for a few days.
Hall C--- Pass 1, QWeak from July 4 to August 13, 2010
· Study Hall C Song Sheet and familiarize yourself with the beam line components.

· Hall C will be down for about 10 day to install its targets.
· Commissioning of Qweak went well so far. Issues were identified and are being addressed.

· Over the weekend beam was steered to the hall dump in straight ahead mode and low CW beam (less than 1 nA) was run. The hall will be down for a few days due to installation work. 
